Author's response to reviews

Title: Identification of a novel de novo mutation in the NIPBL gene in an Iranian patient with Cornelia de Lange Syndrome: a case report

Authors:

Hamid Galehdari (galehdari187@yahoo.com)
Roya Monajemzadeh (Monajemzadeh@gmail.com)
Gholamreza Mohammadian (Mohammadian2000@yahoo.com)
Mohammad Pedram (m_Pedram_2007@yahoo.com)
Habibolah Nazem (Hnazem@gmail.com)

Version: 5 Date: 19 July 2010

Author's response to reviews:

Dear editor of the JMCR,

Following points have been corrected:
1. Abstract;
   a. line 8 - length was changed to height
2. Introduction;
   a. Line 4; estimated prevalence was changed to 1.6 - 2.2 per 100,000 births.
   b. Line 14; ref 7 was changed to ref 8
3. Case presentation;
   a. line 8; cryptorchidism instead of criptoychridism
   b. line 13; length was changed to hight
4. competing interests;
   a. 'I declare that I have no competing interests'
5. References;
   a. have been checked and corrected.

H. Galehdari
Corresponding author